Author's response to reviews

Title: Findings from the SASA! Study: a cluster randomised controlled trial to assess the impact of a community mobilisation intervention to prevent violence against women and reduce HIV risk in Kampala, Uganda

Authors:

Tanya Abramsky (tanya.abramsky@lshtm.ac.uk)
Karen Devries (karen.devries@lshtm.ac.uk)
Ligia Kiss (ligia.kiss@lshtm.ac.uk)
Janet Nakuti (janet@raisingvoices.org)
Nambusi Kyegombe (nambusi.kyegombe@lshtm.ac.uk)
Elizabeth Starmann (elizabeth.starmann@lshtm.ac.uk)
Bonnie Cundill (bonnie.cundill@lshtm.ac.uk)
Leilani Francisco (leilani.francisco@lshtm.ac.uk)
Dan Kaye (dankkaye@yahoo.com)
Tina Musuya (tmusuya@raisingvoices.org)
Lori Michau (lori.michau@raisingvoices.org)
Charlotte Watts (charlotte.watts@lshtm.ac.uk)

Version: 5 Date: 2 July 2014

Author’s response to reviews:

Dear Editor,

I am delighted that our manuscript has been accepted for publication in BMC Medicine. We have uploaded the latest version of the manuscript: "Findings from the SASA! Study: a cluster randomised controlled trial to assess the impact of a community mobilisation intervention to prevent violence against women and reduce HIV risk in Kampala, Uganda".

Please let me know if you require any further information.

Yours sincerely,

Tanya Abramsky
Research Fellow in Epidemiology
Social and Mathematical Epidemiology (SaME)
Gender, Violence & Health Centre
Department of Global Health and Development
Tel +44 (0207 927 8244
E-mail: Tanya.abramsky@lshtm.ac.uk